Author's response to reviews

Title: Cytotoxicity of VEGF121/rGel on Vascular Endothelial Cells Resulting in Inhibition of Angiogenesis is Mediated via VEGFR-2

Authors:

Khalid A Mohamedali (kmohamed@mdanderson.org)
Sophia Ran (sran@siumed.edu)
Candelaria Gomez-Manzano (cmanzano@mdanderson.org)
Latha Ramdas (LRamdas@mdanderson.org)
Jing Xu (Jing.Xu@UTSouthwestern.edu)
Sehoon Kim (SehoonKim@sbcglobal.net)
Lawrence H Cheung (LCheung@mdanderson.org)
Walter N Hittelman (Whittelm@mdanderson.org)
Wei Zhang (WZhang@mdanderson.org)
Johannes Waltenberger (waltenberger@ukmuenster.de)
Philip E Thorpe (Philip.Thorpe@UTSouthwestern.edu)
Michael G Rosenblum (mrosenbl@mdanderson.org)

Version: 8 Date: 15 August 2011

Author's response to reviews:

For some reason the cover letter was uploaded instead of the revised manuscript. This is now being corrected.

Thank you
Khalid